Author's response to reviews

Title: Heteropterys tomentosa (A. Juss.) infusion counteracts Cyclosporin A side effects on the ventral prostate

Authors:

Karine M Freitas (karinemfreitas@gmail.com)
Juliana C Monteiro (julianacmonteiro@gmail.com)
Marcos LM Gomes (marcoslucca@yahoo.com.br)
Sebastião R Taboga (taboga@ibilce.unesp.br)
Heidi Dolder (heidi@unicamp.br)

Version: 4 Date: 21 January 2013

Author's response to reviews: see over
Dear Editor,

We would like to resubmit the corrected version of the manuscript entitled “Heteropterys tomentosa (A. Juss.) infusion counteracts Cyclosporin A side effects on the ventral prostate”. The only suggestion made by the referees was to correct the cited references. We did as suggested by the referee and also according to that observed for recent papers published by the journal.

Sincerely,

Karine Moura de Freitas
ANSWER TO THE REVIEWERS

Reviewer's report
Title: *Heteropterys tomentosa* (A. Juss.) infusion counteracts Cyclosporin A side effects on the ventral prostate
Version: 3 Date: 20 December 2012

Reviewer: Gaffari Turk

Reviewer's report:
Cited references in the discussion section should be given as the surname of first author and et al. [reference number], for example; Kurus et al. [12] observed that........ Türk et al. [18] observed that........  **WE CORRECTED THE CITED REFERENCES**

Level of interest: An article whose findings are important to those with closely related research interests

Quality of written English: Acceptable

Statistical review: No, the manuscript does not need to be seen by a statistician.

Declaration of competing interests: 'I declare that I have no competing interests'

Reviewer's report
Title: *Heteropterys tomentosa* (A. Juss.) infusion counteracts Cyclosporin A side effects on the ventral prostate
Version: 3 Date: 2 January 2013

Reviewer: Katarina Hostanska

Reviewer's report:
Authors responded to all suggestions by reviewers satisfactorily.
Corrected manuscript at present form is acceptable to be published.
Level of interest: An article whose findings are important to those with closely related research interests

Quality of written English: Acceptable

Statistical review: No, the manuscript does not need to be seen by a statistician.

Declaration of competing interests: I declare that I have no competing interests